Comments on Assignment #1

Question #1: Remember that this course is about modelling, thus the format of your answers is very important. You should have a problem statement, model, critique, etc (as per the class handout) in your solution. A lot of people who answered the question correctly lost 10 marks because of the poor format of their answer. Those who didn’t answer the question correctly usually failed to do so because they did not have a sufficient number of variables to portray the problem correctly. Remember, after you have completed your model you should test it to see if it makes sense!

Question #2: You received 10 marks if you tried this question and 0 if you did not. See the solutions to check if your answer is correct.

Question #3: All though this was a numerical question, you should still be following the format that was indicated in the course handout. Remember that you can find a lot of the necessary information for sensitivity analysis in the final tableau/dictionary. A lot of people were cranking out unnecessary information.